Author’s response to reviews

Title: Fertility intentions among HIV positive women aged 18-49 years in Addis Ababa Ethiopia: a cross sectional study Hussen Mekonnen Asfaw * 1; Fikre Enquselassie Gashe 2 1, 2 Department of preventive medicine, School of Public Health, Addis Ababa University, Ethiopia Email: Hussen Mekonnen*:- hussen_mk@yahoo.com; Fikre Enquselassie: -fikreens@yahoo.com
*Corresponding author

Authors:

Hussen Mekonnen Asfaw (hussen_mk@yahoo.com)
Fikre Enquselassie Gashe (fikreens@yahoo.com)

Version: 2 Date: 6 January 2014

Author’s response to reviews: see over
Dear BMC Reproductive Health

First of all I would like to say thank you very much indeed for your concern and valuable comment at this stage.

Please find enclosed revised/ manuscript entitled “Fertility intentions among HIV positive women aged 18-49 years in Addis Ababa, Ethiopia: a cross-sectional survey”. Submitted to BMC Reproductive Health, (Section general) on January 3, 2013.

Based on the comments the following changes were made from previous version

1. **Title part**: authors, authors’ affiliation and corresponding author detail added with the title page.
2. **Abstract part**:
   i. Background part the context totally changed; objectives merged with background
   ii. Methodology two lines were deleted just to minimize sentences
   iii. Conclusion part modification was done

The co-author has agreed with the content of revised manuscript. Once again we certify that the submission/revised version of our manuscript is an original work and is not under review at any other publication.

The finding on this work is important input for resource poor settings to support the undergoing efforts in fighting HIV/AIDS and unplanned pregnancies among the highly affected group of the community (women). Furthermore, it will contribute for the national and international public health decisions by learning contextual factors from different perspectives.

We believe that our findings could be of interest to the readers of BMC Reproductive Health Journals.

Sincerely yours,

Hussen Mekonnen, Fikre Enquselassie

Corresponding author: Hussen Mekonnen, department of preventive medicine, School of Public Health, Addis Ababa University, Ethiopia,

E-mail: hussen_mk@yahoo.com

Cell phone: +251911631634